Author's response to reviews

Title: A pragmatic cluster randomized trial evaluating the impact of a community pharmacy intervention on statin adherence: rationale and design of the Community Pharmacy Assisting in Total Cardiovascular Health (CPATCH) study

Authors:

Charity D Evans (charity.evans@usask.ca)
Dean T Eurich (deurich@ualberta.ca)
Jeff G Taylor (jeff.taylor@usask.ca)
Alfred J Remillard (aj.remillard@usask.ca)
Yvonne M Shevchuk (yvonne.shevchuk@usask.ca)
David F Blackburn (d.blackburn@usask.ca)

Version: 2 Date: 23 June 2010

Author's response to reviews:

Greetings, we have removed the last section in the manuscript (Conclusion) and embedded it as the last section of the discussion. No further requests were made.

thanks for this